Author's response to reviews

Title: The impact of music therapy on managing neuropsychiatric symptoms for people with dementia and their carers: a randomised controlled feasibility study

Authors:

Ming Hung Hsu (ming.hunghsu@mha.org.uk)
Rosamund Flowerdew (ros.flowerfew@mha.org.uk)
Michael Parker (michael.parker@anglia.ac.uk)
Jörg Fachner (jorg.fachner@anglia.ac.uk)
Helen Odell-Miller (helen.odell-miller@anglia.ac.uk)

Version: 2Date: 30 December 2014

Author's response to reviews:

1. Line numbering and page numbers are added to the manuscript.
2. Full authors' details are added to the title page.